Author's response to reviews

Title: Phellodendron and Citrus extracts benefit cardiovascular health in osteoarthritis patients: a double-blind, placebo-controlled pilot study

Authors:

Julius Oben (juliusoben@hotmail.com)
Ebangha Enonchong (juliusoben@hotmail.com)
Shil Kothari (shilkothari@hotmail.com)
Walter Chambliss (chamblissw@aol.com)
Robert Garrison (ddolnick@nextpharmaceuticals.com)
Deanne Dolnick (ddolnick@nextpharmaceuticals.com)

Version: 4 Date: 16 May 2008

Author's response to reviews: see over
Dear Nutrition Journal,

Thank you for your e-mail dated May 16, 2008. We are replying to your requested formatting changes.

1). Affiliations & Please include the country for each affiliation.
Done

2) Table citation & Tables should be cited in the text in consecutive order starting from Table 1 then Table 2, etc. Please reorder the Table citation so that they are cited in consecutive order.
Done

Sincerely,
Deanne Dolnick